Author's response to reviews

Title: A quantitative survey of intern's knowledge of communication skills: An Iranian exploration

Authors:

Mohsen Tavakol (texmt7@nottingham.ac.uk)
Sima Torabi (lgxst1@nottingham.ac.uk)
Owen D Lyne (O.D.Lyne@kent.ac.uk)
Ali A Zeinaloo (zeinaloo@tums.ac.ir)

Version: 5 Date: 4 February 2005

Author's response to reviews: see over
Author's response to reviews

Title: A quantitative survey of intern’s knowledge of communication skills: an Iranian exploration

Authors:
Mohsen Tavakol: texmt7@nottingham.ac.uk
Sima Torabi: lgxst1@nottingham.ac.uk
Owen D. Lyne: O.D.Lyne@kent.ac.uk
Ali A. Zeinaloo: zeinallo@tums.ac.ir

Version: 2 Date: 4 February 2005

Authors’ response to reviews:

To the editor

We have made the following revisions according to the reviewer's report:
1. All minor essential revisions by the reviewer.
2. In addition, we have completed all the formatting changes requested.
3. Some casual language has been eliminated.
4. We have checked our manuscript carefully for any typing errors.